Author's response to reviews

**Title:** Drug injecting and HIV risk among injecting drug users in Hai Phong, Vietnam: A qualitative analysis

**Authors:**

Tanvir Ahmed (tanvir.ahmed@griffithuni.edu.au)
Thanh Nguyen Long (longvaac@gmail.com)
Phan Thi Huong (huongphanmoh@gmail.com)
Donald Stewart (donald.stewart@griffith.edu.au)

**Version:** 2
**Date:** 15 August 2014

**Author's response to reviews:**

1. inserted page numbers
2. inserted line numbers in the main manuscript